Author's response to reviews

Title: The cost effectiveness of an early transition from hospital to nursing home for stroke patients: Design of a comparative study

Authors:

Ron WH Heijnen (R.heijnen@mumc.nl)
Silvia MAA Evers (S.Evers@BEOZ.unimaas.nl)
Trudy DEM van der Weijden (Trudy.vanderWeijden@hag.unimaas.nl)
Martien Limburg (martienlimburg@gmail.com)
Jos MGA Schols (Jos.Schols@hag.unimaas.nl)

Version: 4 Date: 17 May 2010

Author's response to reviews: see over
Dear Editor,

Here are the additional changes that are required:

1. The numbering of references was changed.
2. References 26-27 were altered.

We could find no problem with finding reference 25?

Kind regards,
Ron Heijnen